02/17/2020 2pm – EFOG Meeting Notes
Present: Nena, Dean, Ashley, John, Steve, Shane, Wendy, Clay, Liz

Nena
· Please see this week’s agenda for the action items from last week’s meeting
· At Ops there was a report on the current study
· Nena will send report to the group
· The IOC issue from over the weekend may have been a power issue
· See the agenda for upcoming safety walk throughs, routine rad surveys, and TYSSR
· Everyone please look at your training to see what you are expired in
Events from last week
· 17BM had several warnings.  Solenoid changed out and that seems to have solved the problem.  Clay put in the CCWP and Mark put in the work request.
· 911 for a cell phone overheating at 9ID.  911 responded, waste management eventually took the phone away for disposal; the user and waste management have each other’s contact information incase future correspondence is necessary
· 18ID undulator gap was drifting.  The beamline is running at a smaller gap this run.  John Grimmer was contacted and responded that this just happens and the resolution is either to fix manually or to write a new script.
· Ashley received a call from the MCR at 3:45am on Sunday morning to give her a heads up that a beamline (not specified) was having shutter problems but nothing further was communicated from the MCR after the initial call.
· MCR needs to provide more information
· Per John, Sunday morning the MCR said 5BM had a shutter issue but then it was resolved an hour later
· The log book should have been more detailed
· There was an 8.5 beam loss on Saturday morning.  Several beamline personnel were not happy with the email notifications that were sent out.  This was most likely due to a power pump and when the beam goes back up there were some issues
· John and Clay to email Nena with a recap
· Per John, as soon as beam came back on Saturday, 22 had no beam.  Vacuum Permit EPS was able to be reset but the EPS Shutter permit couldn’t be reset.  PSS was showing EPS by EPCIS was not showing that.  Greg Banks and Nick Demanti were called.  They couldn’t find anything.  John Krass came in and broke vacuum.  It was eventually determined that the UPS for the BLEPS was broken.  Plugged it in directly and bypassed UPS and everything was ok.  22 now back to their UPS.
· 14IDC had a shutter issues but rebooting solved the problem.
· 26ID did not have any beam.  Mirror was in the way.  Issue resolved.
Liz
· Sector 9 walk through went well.  No major issues
· Mike Fries is in the process of buying EFOG a new Be cart and vacuum for the cart.
Clay
· 10ADM-610 trip due to users bumping the chain.  Clay talked with the users about this and also about how they should not ignore alarms when they go off
· Clay tied the chain out of the way.
Wendy
· Two rad experiments start on Tuesday 2/18 at 1IDE and 13IDD.  1IDE requires a dosimeter to be worn if entering the hutch.  See rad experiment update email for further information regarding these two experiments.
· New HP tech started today, Kelly.
Shane
· Purdue material currently at 32.  Will move to Sector 2 when needed.  
· Shane will check in with group and send email update to EFOG
· Put in a vector request if you have items you want to excess that need to be sanitized.
· Shane to excess small computer Patti used to use to take meeting notes and the same small laptop that Clay has.
· Shane please forward email to Nena regarding key request issue.
Steve
· Slow Valve fault at 20BM; Banks doing coding update so this was a non-issue
· Alex D at 32 had a CO2 cylinder than was corroded on the bottom and got a hold of Mike Fries regarding moving the cylinder.  Mike said it was ok to move to the truck lock.  Mike to notify Airgas.
John
· 7ID AR still needs to be posted
· Nena to remind Bruno about this
· John had to fill out a conflict of interest form that was showing up as expired.  Now good until April
· Francesco was concerned about some beam motion at 2BM.  A new camera was recently purchased.  Karen said all was operating as normal.
· 14 walk through was good.
· Routine Rad Surveys
· 4ID Thursday
· 6BM Friday – Dean to do
· Next week: 29IDD Wed, 33ID We, 34ID Thursday – John to do
Ashley
· 7ID put up a sign to remind people about their ESAFs.  This was much appreciated.
Dean
· Steve is doing great.
· [bookmark: _GoBack]He found a laser level and will purchase soon.
